Author's response to reviews

Title: Uterotonic use immediately following birth: Using a novel methodology to estimate population coverage in four countries

Authors:

Jim Ricca (Jim.ricca@jhpiego.org)
Vikas Dwivedi (vdwivedi@jsi.com)
John Varallo (John.varallo@jhpiego.org)
Gajendra Singh (Gajendra.Singh@jhpiego.org)
Suranjeen P Pallipamula (Suranjeen.Pallipamula@jhpiego.org)
Nazir Amade (naibrahimo@yahoo.com.br)
Maria de Luz Vaz (Maria.vaz@jhpiego.org)
Dustan Bishanga (dustan.bishanga@jhpiego.org)
Marya Plotkin (Marya.plotkin@jhpiego.org)
Bushra Al-Makaleh (Bushra.Al-Makaleh@jhpiego.org)
Stephanie Suhowatsky (Stephanie.suhowsksy@jhpiego.org)
Jeffrey M Smith (Jeffrey.Smith@jhpiego.org)

Version: 4 Date: 12 December 2014

Author's response to reviews:

Included all tables in the main manuscript document. this was the only requested change.